Author's response to reviews

Title: Ethanol extract of propolis protects macrophages from oxidized low density lipoprotein-induced apoptosis by inhibiting endoplasmic reticulum stress-C/EBP homologous protein pathway

Authors:

Hua Tian (liutianfangyu@163.com)
Hongwei Sun (hongweisunyst@126.com)
Jiajun Zhang (gegewv_008@163.com)
Xiaowei Zhang (drgingfeng@163.com)
Li Zhao (zhaoli20131234@126.com)
Shoudong Guo (gsd20060501@163.com)
Yanyan Li (1228441486@qq.com)
Peng Jiao (jiaopeng196@163.com)
Hao Wang (tywanghao_2005@163.com)
Shucun Qin (shucunqin@hotmail.com)
Shutong Yao (yst228@126.com)

Version: 3  Date: 10 March 2015

Author's response to reviews: see over
Ethanol extract of propolis protects macrophages from oxidized low density lipoprotein-induced apoptosis by inhibiting endoplasmic reticulum stress-C/EBP homologous protein pathway (MS: 1651053095158727)

Dear Editor,

The use of human blood and protocol of human LDL isolation were approved by the Ethics Committee of Taishan Medical University, and all blood donors provided informed consent for the use of their blood. The information has been added in the “Isolation and oxidation of LDL” of Methods section.

Yours sincerely,

Hua Tian